Author's response to reviews

Title: Weight loss intervention for individuals with high internal disinhibition: Design of the Acceptance Based Behavioral Intervention (ABBI) randomized controlled trial.

Authors:

Jason Lillis (jasonlillis22@gmail.com)
Heather M Niemeier (niemeieh@uww.edu)
Kathryn M Ross (kross3@lifespan.org)
John G Thomas (jthomas4@lifespan.org)
Tricia Leahey (tricia.leahey@unconn.edu)
Jessica Unick (junick@lifespan.org)
Kathleen E Kendra (kkendra10@gmail.com)
Rena R Wing (rwing@lifespan.org)

Version: 4 Date: 19 May 2015

Author's response to reviews: see over
May 19th, 2015

*BMC Psychology*
Dr. Anna Clark, Executive Editor

To Whom it May Concern:

Thank you for considering our manuscript titled, “Weight loss intervention for individuals with high internal disinhibition: Design of the Acceptance Based Behavioral Intervention (ABBI) randomized controlled trial” for publication in *BMC Psychology*. This Study Protocol presents the design of a randomized controlled trial funded by NIDDK. The manuscript was originally submitted on April 1st, 2015. We have made the following changes the manuscript as requested:

- The title page now includes all author affiliations, email addresses, and country.
- The “Funding Source” section has been renamed as “Acknowledgement”
- All of the tables have been moved to after the references
- Figure 1 has been reformatted and uploaded as a pdf

Consistent with the previous submission, all of the following still apply:

1. Ethical and funding approval documentation have been submitted to the BMC editorial office via email attachments.

2. This study is funded by the National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK) and has been peer-reviewed by the funding agency.

3. This study is ongoing; we are in a no-cost extension and will continue to collect data throughout the year.

4. There are no publications or submitted manuscripts using the data from this protocol at this time.

5. The authors have no competing financial or non-financial competing interests to declare. We have added a section indicating this on the manuscript.

6. The paper authors are Jason Lillis, Heather Niemeier, Kathryn Ross, Graham Thomas, Tricia Leahey, Jessica Unick, Kathleen Kendra and Rena R. Wing. My colleagues and I do not have any interests that might be interpreted as influencing the research, and APA ethical standards were followed in the conduct of this study. All authors have had access to the data, have been personally and actively involved in substantive work leading to the report, and will hold themselves jointly and individually responsible for its content. All authors approved the final manuscript. We have added a section on “authors’ contributions” to the manuscript.
7. No sponsor reviewed the manuscript text prior to submission. All methods were approved by IRB.

8. The clinicaltrials.gov registration date has been added to the abstract of the manuscript.

I will be serving as the corresponding author for this manuscript. Thank you very much for your consideration.

Sincerely,

Jason Lillis, Ph.D.
Assistant Professor (Research)
Brown Alpert Medical School/ The Miriam Hospital
Weight Control and Diabetes Research Center
196 Richmond Street
Providence, RI 02903
Phone: (401) 474-6108
Fax: (401) 793-8944
jasonlillis22@gmail.com